Author's response to reviews

Title: The Impact of Statins on Health Services Utilization and Mortality in Older Adults Discharged from Hospital with Ischemic Heart Disease: a cohort study.

Authors:

Charmaine A Cooke (charmaine.cooke@dal.ca)
Susan A Kirkland (susan.kirkland@dal.ca)
Ingrid S Sketris (ingrid.sketris@dal.ca)
Cox Jafna (jafna.cox@dal.ca)

Version: 4 Date: 20 October 2009

Author's response to reviews: see over
October 20, 2009

Editorial Office, BioMed Central Ltd
Re: Manuscript number 1358258142239674

To the Editorial Office,

I am submitting a final version of my manuscript entitled ‘The impact of statins on health services utilization and mortality in older adults discharged from hospital with ischemic heart disease: a cohort study’.

I have made all of the formatting changes requested as per your email dated October 13, 2009. Thank you for the opportunity to publish my manuscript.

Sincerely,

Charmaine Cooke, BSc (Pharm), MSc (Community Health and Epidemiology)
Project Coordinator, Population Health Research Unit, Dalhousie University